\documentstyle{article}

\pagestyle{empty}
\thispagestyle{empty}
\begin{document}


\begin{center}
{\bf EECS 225d: Audio Signal Processing by Humans and Machines} \\
\vspace{0.6cm}

\begin{Large}
{\bf  Problem Set 3} \\
{\bf Due in class Wed Feb 19, 1997} \\
\end{Large}

\end{center}

\vspace {0.1cm}

{\sc Reading:} Chapters 8-9

\vspace {0.2cm}

\begin{enumerate}
\item
Chapter 8, Exercise 1

\item
Chapter 8, Exercise 2

\item
Chapter 8, Exercise 3

\item
Chapter 8, Exercise 4

\item
Chapter 9, Exercise 1

\item
Chapter 9, Exercise 2

For this problem, you can assume that you are given a codebook of the 16 
reference height / weight pairs.

\item
Chapter 9, Exercise 3

\end{enumerate}
\end{document}
